Author's response to reviews

Title: Dental Caries Status of Dai Preschool Children in Yunnan, China

Authors:

S N Zhang (s.nan1984@hotmail.com)
Juan Liu (liujuan1966@yahoo.com)
Edward CM Lo (hrdplom@hku.hk)
C H Chu (chchu@hku.hk)

Version: 8 Date: 20 November 2013

Author's response to reviews: see over
The Editor,
BMC Oral Health,
BioMed Central Ltd,
Floor 6, 236 Gray's Inn Road,
London, WC1X 8HL,
United Kingdom.

20 November 2013

Dear Editor,

Re: Submission of manuscript

Thank you for your reviewing of our revised manuscript entitled *Dental Caries Status of Dai Preschool Children in Yunnan, China*. We have revised the manuscript according to your comment:

<table>
<thead>
<tr>
<th>Editor’s comment</th>
<th>Authors response</th>
</tr>
</thead>
<tbody>
<tr>
<td>Caries is referred to as a plural throughout the paper when it should be referred to in the singular.</td>
<td>Agree. Amendments have been made in the manuscript (Page 1, line 10; page 7, line 169; and page 11, line 258 and line 270 in red.)</td>
</tr>
</tbody>
</table>

We sincerely hope you will accept our revised manuscript.

Thank you for your attention.

Yours sincerely,

Dr C H Chu
Corresponding author